

Teenage childbearing and school dropout in a sample of 18,791 single mothers in Cameroon

Joëlle Sobngwi-Tambekou, Marthe Tsague-Agnoux, Léopold Fezeu, Flavien Ndonko

▶ To cite this version:

Joëlle Sobngwi-Tambekou, Marthe Tsague-Agnoux, Léopold Fezeu, Flavien Ndonko. Teenage child-bearing and school dropout in a sample of 18,791 single mothers in Cameroon. Reproductive Health, $2022,\ 19\ (1),\ pp.10.\ 10.1186/s12978-021-01323-4$. hal-03622607

HAL Id: hal-03622607 https://cnam.hal.science/hal-03622607

Submitted on 29 Mar 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



RESEARCH Open Access

Teenage childbearing and school dropout in a sample of 18,791 single mothers in Cameroon

Grossesses précoces et abandon scolaire dans un échantillon de 18,791 mères célibataires au Cameroun

Joëlle L. Sobngwi-Tambekou^{1,2,3*}, Marthe Tsague-Agnoux², Léopold K. Fezeu⁴ and Flavien Ndonko⁵

Abstract

Background: Adolescent childbearing increases the risk of adverse health and social consequences including school dropout (SDO). However, it remains unclear why some teenage mothers drop out of school and others do not, especially in sub-Saharan Africa settings. We aimed to investigate the background and behavioral characteristics of single mothers, associated with school dropout in a sample of 18,791 Cameroonian girls, who had their first child during adolescence.

Methods: We used data from a national registry of single mothers, collected during the years 2005–2008 and 2010–2011. Both bivariate analysis and logistic binary regression models were used to explore the relationship between adolescence motherhood and SDO controlling for a range of socio-economic, family, sexual and health seeking behavior characteristics.

Results: Among the 18,791 single mothers, 41.6% had dropped out of school because of pregnancy. The multivariable regression model showed that SDO was more common in those who were evicted from their parental home (aOR: 1.85; 95% CI: 1.69–2.04), those who declared having other single mothers in their family (aOR: 1.16; 95% CI 1.08–1.25) and in mothers who had their first child before 15. Using modern contraceptive methods, having declared no sexual partner during the last year and having less than 2 children were associated with a reduced likelihood of school dropout.

Conclusions: Strong social support is essential to ensure school continuity in this vulnerable population. Dropping out of school may put the teenage mother more at risk of unsafe health behaviour and new pregnancies.

Keywords: Teenage pregnancy, Single mothers, School dropout, Family support, School continuity

¹ RSD Institute, Rue de l'Université, Po Box 7535, Yaoundé, Cameroon Full list of author information is available at the end of the article



^{*}Correspondence: sobngwijoelle@gmail.com

Résumé

Contexte: La maternité chez les adolescentes augmente le risque d'évènements sanitaires et sociaux néfastes, notamment l'abandon scolaire (SDO). Cependant, en Afrique subsaharienne, les raisons pour lesquelles certaines mères adolescentes abandonnent l'école tandis que d'autres réussissent à poursuivre leurs études ne sont pas toutes connues. Dans cet article, nous avons étudié les caractéristiques comportementales et sociodémographiques des mères célibataires, associées à l'abandon scolaire dans un échantillon de près de 18.791 filles camerounaises, qui ont eu leur premier enfant pendant l'adolescence.

Méthodes: Nous avons utilisé les données d'un registre national des mères célibataires, collectées au cours des années 2005-2008 et 2010-2011. Une analyse bivariée et des modèles de régression binaire logistique ont été utilisés pour explorer la relation entre la maternité à l'adolescence et l'abandon scolaire, en contrôlant une série de caractéristiques socio-économiques, familiales, sexuelles et sanitaires.

Résultats: Parmi les 18 791 mères célibataires, 41,6% avaient abandonné l'école en raison d'une grossesse. Le modèle de régression multivariable a montré que l'abandon scolaire était plus fréquent chez celles qui avaient été expulsées du domicile parental (ORa: 1,85; IC 95%: 1,69-2,04), chez celles qui déclaraient avoir d'autres mères célibataires dans leur famille (ORa: 1,16; IC 95%: 1,08-1,25) et chez les mères qui avaient eu leur premier enfant avant 15 ans. L'utilisation de méthodes contraceptives modernes, le fait de n'avoir déclaré aucun partenaire sexuel au cours de la dernière année et d'avoir moins de deux enfants étaient associés à une probabilité réduite d'abandon scolaire.

Conclusions: Un soutien social fort est essentiel pour assurer la continuité scolaire dans cette population vulnérable. L'abandon de l'école peut exposer la mère adolescente à un risque accru de comportements sanitaires dangereux et de nouvelles grossesses.

Mots clés: Grossesse adolescente, Mères célibataires, Abandon scolaire, Soutien familial, Continuité scolaire

Plain language summary

There is compelling, worldwide evidence that pregnancy and birth during teenage years are significant contributors to high school dropout rates, especially in developing countries. Research has also shown that education continuity of teenage mothers can lessen the long term negative social consequences of teenage pregnancies and childbearing. In this study, we investigated the factors associated with school dropout in a sample of Cameroonian teenage mothers.

The information collected included socio-economic, family, sexual characteristics, and health-seeking behavior.

Among the 18,791 single mothers, 41.6% had dropped out of school because of pregnancy. School dropout was more common in those who were evicted from their parental home, those who declared having other single mothers in their family and in mothers who had their 1st child before 15. Using modern contraceptive methods, having declared no sexual partner during the last year and having less than 2 children were associated with a reduced likelihood of school dropout.

In conclusion, strong social support is essential to ensure school continuity in this vulnerable population. Dropping out of school may put the teenage mother more at risk of unsafe health behaviour and new pregnancies.

Background

Despite all the policies and interventions that have been implemented throughout the world to prevent teenage pregnancies, they continue to be a challenge, especially in developing countries. According to the United Nations Population Fund, approximately 12 million girls aged 15 to 19 years and at least 777,000 girls under 15 years give birth each year in developing regions [1]. While the adolescent birth rate (ABR) that is, the number of births per 1000 women aged 15 to 19, has fallen in most countries, it is still rising in low-income settings. In sub-Saharan Africa, the ABR rose from 98 between 2010 and 2015, to 104 between 2015 and 2020 [2, 3]. According to the

World Health Organization (WHO), adolescence is the period of life between 10 and 19 years old [4].

Research has shown that teenage pregnancies increases the risk of adverse health and social consequences for both the mother [5–7] and her child [8–11]. Children of teenage mothers are much more likely to have lower health assessment scores, lower academic achievement and higher risk of not completing high school during adolescence, compared to those of women that have their first birth at age 20 to 21 [12].

There is compelling, worldwide evidence that pregnancy and birth during teenage years are significant contributors to high school dropout rates [13–22].

A population-based study carried out in Cameroon revealed that pregnancy is the second cause of school dropout in girls with 16.9% dropout rates, after financial problems [23]. School dropout is therefore an additional barrier these adolescent girls who are already severely handicapped by parenthood, must face to overcome the longer-term repercussions of childbearing. Some studies have argued that the long-term negative consequences of early birth on a mother's social status and child development could be lessened by continuing education, after an adolescent birth, suggesting that teenage single mothers' access to education should be strongly encouraged. Educational continuity may reduce the risk of subsequent pregnancies, enable teenage mothers to access more lucrative jobs and ensure improved cognitive and behavioral outcomes among children [9, 24-26]. However, the reasons why some teenage mothers drop out of school and others do not, are still unclear especially in sub-Saharan African settings.

Given the importance of education in girls, we aimed to investigate the background and behavioral characteristics of single mothers, associated with school dropout in a sample of 18,791 Cameroonian girls, who had their first child during adolescence. The purpose of this study is to drive the development of strategies to prevent school dropout and promote continuing education within this vulnerable population.

Methods

Study population

We used data from a national registry of single mothers who had their first child during adolescence, involved in the RENATA (Réseau National des Tantines) project. RENATA is the French acronym for the National Network of Aunties Associations, an aunty being a teenage mother trained in sexual and reproductive health and rights of adolescents.

This network consists of more than 20,000 single mothers trained as "Aunties" in about 305 community-based associations in Cameroon. RENATA is supported by a multidisciplinary team of psychologists, anthropologists and communication specialists. The recording of the profile of teenage mothers began in 2003 and this data was collected before each training session of a new Aunties' association in the community and continued systematically until 2012, nationwide. The main aim of the RENATA network is to ensure psychosocial support to single mothers through awareness about sexual abuse, violence against girls and harmful practices such as breast ironing and other forms of injustice against the girl child. The network's activities also involve counseling in sexual and reproductive health, promotion and support of initiatives to empower and protect teenage mothers. The RENATA network received initial funding from the German cooperation through the German-Cameroon Health Programme implemented by the Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ) [27].

Sampling technique and sample size

This is a retrospective review of the RENATA's network database. Every girl who gave birth to one or more children between 10 and 19 years and was not married at the time of birth of the child was considered a teenage mother.

The inclusion criteria were: having given birth to at least one child between 10 and 19 years, being single and younger than 30 years at the time of the survey, 30-year-old was the age limit to be eligible to be part of the network.

Ten surveys were conducted between 2003 and 2012 and 23,723 adolescent mothers included. For this review, we excluded data collected before the year 2005 which did not include information on the participants' reported sexual behavior like: the number of sexual partners during the last 12 months, history of rape, use of condom during the last sexual intercourse and use of a modern contraceptive method. Data from the years 2009 and 2012 were also excluded because of the small number of individuals included those years. With these restrictions, the resulting sample that was used for analyses was 18,791 single mothers recruited between 2005–2008 and 2010–2011.

Dependent variable

The dependent variable was school dropout due to unplanned pregnancy or childbirth.

Independent variables

The independent variables were the socioeconomic characteristics of the single mother: age at 1st childbirth in 2 groups (10–14 years and 15–19 years), the current age of the single mother, her educational level (vocational school, primary, secondary and higher), the number of children she has, the total amount of expenses during the last month and the region of residence (sahel, forest, coastal and grassfield region) that refers to the socio-anthropological areas of the country in which she lives. Another set of independent variables was the family background. The participants were asked if they have been sent away from their family house because of the pregnancy, if there were other cases of unplanned/early pregnancies within their family, what was the profession of the father of their last child and if the father was taking care of the child. The reported sexual and health seeking behavior of the teenage mothers were: age at 1st sexual intercourse, number of sexual partners during the

last 12 months, history of rape, history of induced abortion, HIV screening and the use of modern contraceptive methods. These variables were assessed, based on participants' self-reports.

Statistical analyses

We used univariate analyses to describe the teenage mother's population (proportions, percentiles, means \pm standard deviation (SD)). Bivariate analyses were conducted to compare single mother school dropouts to non-school dropouts socioeconomic, sexual behavior and family background (Student t tests and X^2 tests).

A multivariable binary logistic regression was conducted using the stepwise forward modelling to estimate the odds ratio (OR) of school dropout in teenage mothers. The model was adjusted for the year of inclusion in the study to address potential effect of the year of inclusion, as our preliminary analyses found a statistically significant correlation between school dropout and the year of registration to the network. The model included all control variables simultaneously, except the current age of the teenage mother. It allowed estimating the likelihood of school dropout according to the different factors associated to school dropout. Odd Ratios (OR) and adjusted ORs with their 95% confidence intervals (95% CI) were computed, and all indicators were estimated using the statistical software SPSS® version 16 (SPSS, Chicago, IL).

Results

Among the 18,791 single mothers, 42.3% had dropped out of school because of unplanned pregnancy. The main demographic, family and sexual behavioral characteristics of the single mothers are presented in Table 1. The mean \pm SD age at the time of the study was 20.5 ± 2.9 years. Overall, 33.4% of the mothers declared having at least two children. Over 60% of the single mothers had completed at least secondary school with only 0.8% at university at the time of the baseline survey. More than half (51.3%) of the mothers admitted that they had lost years of schooling. The mean age at 1st birth \pm SD was 17.2 ± 1.5 in non-dropouts, compared to that of school dropout teenage mothers (16.4 \pm 1.6; P < 0.001). Among the dropouts, 16.3% declared having been sent away from the family house because of pregnancy. Over half of the single mothers (57.3%) declared that the father of the most recent child was taking care of his/her needs. More than two-thirds (69.0%) of the participants reported having other family members who gave birth during adolescence. This proportion was higher in dropouts than in non-dropouts (71.5% vs. 67.2%; P<0.000). The median amount (25th-75th percentiles) of expenses of the participants during the last month was 30 (20-60 euros).

The mean \pm SD age at first sexual intercourse was 15.4 ± 1.7 years old. Almost one in five (19.3%) said they had an induced abortion at least once. HIV screening during the last year was performed by only 65.1% of the mothers and only 32.8% declared having used condom during their last sexual intercourse. Participants who reported more than one sexual partner during the last 12 months and those who declared having ever being raped were more likely to drop out from school (8.3% vs. 7.0%; P < 0.001).

Results of the multivariable model are presented in Table 2. School dropout was positively associated with being sent away from the parental home because of pregnancy (aOR: 1.85; 95% CI: 1.69-2.04). Single mothers who declared a primary or vocational school education level at the time of the study, had lower likelihood of SDO than those with high school or University education level (aOR: 0.40; 95% CI 0.37-0.42 and aOR: 0.30; 95% CI 0.23-0.38 respectively). Those who declared having single mothers in their family were more likely to dropout. Compared to those who had their first child after 14 years old, mothers who had their first child before had a 50% increased likelihood of school dropout. School dropout was more likely found in participants whose last child's father was not providing for (aOR: 1.20; 95% CI 1.13–1.28) and in those who reported having never done an HIV screening test (aOR: 1.15; 95% CI 1.08-1.23). The likelihood of SDO increased with the number of partners or the number of children.

Discussion

The results of this data review on the largest network of single in Cameroon showed that more than half of the girls (54%) had to drop out of school for at least some time because of pregnancy and childbirth, and 77% of them (46.1% of all mothers) dropped out of school permanently. Early age (less than 14 years) at 1st childbirth and factors like having being sent away from home because of pregnancy, the father's child not providing for him/her, existence of other single teenage mothers in the family and history of rape were associated with increase odds of school dropout, while responsible personal health behaviors such as having only one or no sexual partner during the previous year, the use of modern contraceptive methods and voluntary HIV testing were associated with decreased odds of school dropout. These associations remained statistically significant after controlling for potential confounding factors related to the teenage sociodemographic, family and sexual behavioral characteristics.

Table 1 Association between sociodemographic, family and reproductive characteristics, and school dropout among single mothers in Cameroun

Variables	Total (N = 18,791)	Dropouts (n = 7941)	Non-dropouts $(n = 10,850)$	p-value
Sociodemographic characteristics				
Mean \pm SD age at the time of the study (years)	20.5 ± 3.0	20.4 ± 3.0	20.6 ± 2.9	0.001
Mean ± SD age at 1st child's birth of the (years)	17.1 ± 1.5	16.7 ± 1.6	17.1 ± 1.5	0.001
Age at 1st child's birth (%)				
10–14	1244 (6.6)	654 (8.2)	590 (5.4)	0.001
15–19	17,547 (93.4)	7287 (91.8)	10,260 (94.6)	
Level of education (%)				
Vocational school	395 (2.1)	99 (1.2)	296 (2.7)	0.001
Primary school	6850 (36.5)	2021 (25.5)	4829 (44.5)	
High school/University	11,546 (61.4)	5821 (73.3)	5725 (52.8)	
Socioantropological region (%)				0.001
Coast	4395 (23.4)	1553 (19.6)	2842 (26.2)	
Forest	9188 (48.9)	4670 (58.8)	4518 (41.6)	
Grassfield	4565 (24.3)	1498 (18.9)	3067 (28.3)	
Sahel	643 (3.4)	220 (2.8)	423 (3.9)	
Number of children (%)				0.001
1	12,528 (66.7)	5014 (63.1)	7514 (69.3)	
2–3	5722 (30.5)	2694 (33.9)	3028 (27.9)	
≥ 4	541 (2.9)	233 (2.9)	308 (2.8)	
Amount of expenses during last month (%)				
< 20 USD	6455 (34.4)	2565 (32.3)	3890 (35.9)	0.001
20-50 USD	6380 (34.0)	2687 (33.8)	3693 (34.0)	
≥ 50 USD	5956 (31.7)	2689 (33.9)	3267 (30.1)	
Family environment				
Sent away from family house (%)				0.001
Yes	2317 (12.3)	1298 (16.3)	1019 (9.4)	
No	16,474 (87.7)	6643 (83.7)	9831 (90.6)	
Father providing for the child needs (%)				0034
Yes	10,765 (57.3)	4478 (56.4)	6287 (57.9)	
No	8026 (42.7)	3463 (43.6)	4563 (42.1)	
Family history of teenage mothers (%)				0.01
Yes	12,970 (69.0)	5678 (71.5)	7292 (67.2)	
No	5821 (31.0)	2263 (28.5)	3558 (32.8)	
Partner's profession (%)				0.001
Student	5018 (26.7)	2527 (31.8)	2491 (23.0)	
Trader	2551 (13.6)	926 (11.7)	1625 (15.0)	
Craftsmen	7623 (40.6)	2929 (36.9)	4694 (43.3)	
Other	801 (4.3)	354 (4.5)	447 (4.1)	
Unemployed	1787 (9.5)	770 (9.7)	1017 (9.4)	
Civil servant	1011 (5.4)	435 (5.5)	576 (5.3)	
Reported sexual behaviour				
Mean ± SD age at 1st sexual intercourse (years)	15.4 ± 1.7	15.2 ± 1.7	15.4 ± 1.7	0.001
Number of sexual partners during the last 12 months (9				0.001
0	1092 (5.8)	429 (5.4)	663 (6.1)	
1	13,390 (71.3)	5456 (68.7)	7934 (73.1)	
≥ 2	4309 (22.9)	2056 (25.9)	2253 (20.8)	

Table 1 (continued)

Variables	Total (N = 18,791)	Dropouts (n = 7941)	Non-dropouts $(n = 10,850)$	p-value
Was raped (%)				
Yes	1418 (7.5)	658 (8.3)	760 (7.0)	0.001
No	17,373 (92.5)	7283 (91.7)	10,090 (93.0)	
Ever had an induced abortion (%)				
Yes	3623 (19.3)	1542 (19.4)	2081 (19.2)	0.682
No	15,168 (80.7)	6399 (80.6)	8769 (80.8)	
Ever done HIV screening test (%)				
Yes	12,229 (65.1)	5087 (64.1)	7142 (62.8)	0.012
No	6562 (34.9)	2854 (35.9)	3708 (34.2)	
Use of a modern contraceptive method (%)				
Yes	7334 (39.0)	3133 (39.5)	4201 (38.7)	0.311
No	11,457 (61.0)	4808 (60.5)	6649 (61.3)	
Use condom during the last sexual intercourse (%)				
Yes	6162 (32.8)	2698 (34.0)	3464 (31.9)	0.003
No	12,629 (67.2)	5243 (66.0)	7386 (68.1)	
Ever had and STD (%)				
Yes	3760 (20.0)	1325 (16.7)	2435 (22.3)	0.001
No	15,031 (80.0)	6616 (83.3)	8415 (77.7)	
Had an ANC visit during pregnancy (%)				
Yes	17,900 (95.3)	7571 (95.3)	10,329 (95.2)	0.65
No	891 (4.7)	370 (4.7)	521 (4.8)	

The rate of school dropout in this study population is higher than that of girls in the national sample of the Multiple Indicators Cluster Survey in Cameroon (16%) [28]. This is understandable given the fact that pregnancy and childbearing are among the main determinants of school dropout in teenage adolescents and our study population is a selection of teenage mothers [29]. Indeed, several studies have highlighted the high prevalence of SDO among teenage mothers [13, 22, 30–34]. Timæus et al. prospectively demonstrated that relationship in a cohort of 673 childless and school teenage girls. Girls who subsequently gave birth had 4.1 times the odds of other girls of dropping out of school [22].

Social and economic vulnerability of teenage mothers play a role in their ability to continue school or not. In this sample, mothers without family support neither from their parents because they were sent away from home nor from the child's father, were more likely of dropping out of school. Similar conclusions were drawn by other authors who showed that, adolescent mothers living with their parents were more likely to stay in school or get back to school promptly after giving birth [22, 32]. This situation is compounded by the discrimination the girls may face from the community and particularly from their peers and teachers in school. Although several African countries have adopted laws and policies that protect the

rights of adolescent girls to stay in school during pregnancy and motherhood, some continue to promote discriminatory and punitive measures against adolescent girls who become pregnant [35]. Pregnant adolescent girls are sometimes stigmatized, booed by their peers and teachers, or even excluded from schools [29, 36]. Even in countries that support the educational continuity of adolescent mothers, few have support structures in schools and communities to assist adolescent mothers who choose to stay in school [35]. SDO thus seems inevitable when teenage mothers are left in the front line of caring for their babies, without affective and financial support from their families and community, to cope with the responsibilities of and adjustment to parenthood. Studies have shown that adequate support from family and partner and positive interactions within the social network were associated with good maternal behavior, life satisfaction in mothers, and school continuity in adolescent mothers [6, 25]. Being deprived of this support in a country where professional support specifically targeted at this vulnerable population is not developed leads to the isolation of the teenage mother, accentuates the precariousness of her economic condition and leaves her without any chance to continue her education. Social support needs to be strengthened around adolescent mothers to encourage educational continuity. In particular,

Table 2 Factors associated to school dropout in 18,791 teenage single mothers recruited between 2005 and 2011, in Cameroon

	n	OR	95% C.I	p-value	ORa	95% C.I	p-value
Age at first childbirth							
10–14 ans	1244	1.11	0.99-1.24	0.07	1.49	1.32-1.69	0.001
15–19 ans	17,547	1			1		
Educational level							
Vocational school	395	0.33	0.27-0.42	0.001	0.30	0.23-0.38	0.001
Primary school	6850	0.42	0.40-0.44	0.001	0.40	0.37-0.42	0.001
High school or University	11,546	1			1		
Sent away from home							
Yes	2317	1.27	1.17-1.38	0.001	1.85	1.69-2.04	0.001
No	16,474	1			1		
Father providing for the child needs							
Yes	10,765	1			1		
No	8026	1.41	1.35-1.47	0.001	1.20	1.13-1.28	0.001
Family history of teenage mothers							
Yes	12,970	1.28	1.23-1.33	0.001	1.16	1.08-1.25	0.001
No	5821	1			1		
Number of sexual partners during the last 12 months							
0	1226	0.65	0.57-0.73	0.001	0.79	0.68-0.91	0.001
1	14 758	0.69	0.66-0.71	0.001	0.86	0.80-0.92	0.001
≥ 2	4713	1			1		
Use condom during the last sexual intercourse							
Yes	6162	1			1		
No	12,629	1.28	1.22-1.35	0.001	1.01	0.94-1.07	0.873
Ever done HIV screening test							
Yes	12,229	1			1		
No	6562	1.41	1.22-1.35	0.001	1.15	1.08-1.23	0.001
Number of children							
1	12,528	0.67	0.64-0.69	0.001	0.72	0.60-0.87	0.007
2–3	5722	0.89	0.84-0.94	0.001	1.03	0.85-1.25	0.409
≥ 4	541	1			1		
Was raped							
Yes		1			1		
No		0.87	0.78-0.96	0.007	0.88	0.78-0.98	0.026

^a Adjusted for every variable presented and year of inclusion in the study, socioanthropological region, partner's profession, amount of expenses during the last month, condom use during the last sexual intercourse, age at first sexual intercourse, ever had an STD

governments must commit to combating discrimination in schools based on pregnancy, in policy and practice, to ensure that there are no legal barriers to keeping teenage mothers in school.

In addition, SDO was associated with a risky sexual and health behaviour. This result is consistent with that of many other studies which stated that school attendance may influence sexual network and increase health-seeking behaviour [37–40]. Indeed, in a context marked by the scarcity of opportunities for professional integration, teenage mothers who leave school are inevitably exposed to unwanted sexual advances from adult males, coercive sexual relations, unequal gender power relations, lack of comprehensive sexuality education, non-use of

contraceptives and inappropriate forms of recreation [41–45]. They are excluded from conventional channels of information on responsible sexuality and reproductive health and have limited access to the basic health services that are usually offered to adolescents in school settings. They thereby miss out on the opportunity to be exposed to all these services that may enable them to ensure good health for themselves and their offspring. Overall, studies have highlighted the poor access of teenage mothers to sexual and reproductive health services [44, 45]. Youth-centered and friendly reproductive health services that can accommodate young people during non-working hours, i.e. before or after school, should be developed within communities and particularly within schools to

improve young people's access to reproductive health information and services.

In this study, 7.5% of the single mothers declared having ever been raped and those ever been raped were more likely to drop out of school. This result is a further indicator of the vulnerability of single mothers and an indication of the urgency of scaling up interventions to raise awareness about sexual abuse, violence, and other harmful practices against children in general and specifically the girl child. The association between early age at first childbirth and SDO highlighted by this study suggests that adolescents and families should be exposed early enough to sexual education to postpone the age at first birth and reduce the risk of SDO. Given the increasing importance of female school participation in sub-Saharan Africa, sexual education should not be limited in schools or health facilities but should be an integrated part of the education given by parents in homes.

This study is the first to present the association between childbearing and school attendance in almost 19,000 single mothers living in a sub-Saharan country. Nevertheless, the study presents some limitations. The generalization of the results to all teenage mothers in the country must be cautious. This sample is mainly constituted of teenage mothers living in precarious socio-economic conditions, there may be differences in educational path among teenage mothers with different socio-economic background. However, the results obtained from this large-scale sample of 18,791 adolescent mothers from all regions of the country may likely reflect the characteristics of teenage mothers from disadvantaged backgrounds, in countries with socioeconomic and cultural characteristics like Cameroon's. Additional limitations of this study included the difficulty controlling for any confounder related to the teenage parents' household (single parents, number of siblings, sibling rank, etc.) because the information was not available in the database. Since this study is cross-sectional, no causal link can be established between school drop-out and the characteristics of adolescent mothers.

Conclusions

The results of this study in 18,791 teenage mothers showed that strong social support is essential to ensure school continuity in this vulnerable population and dropping out of school may put the teenage mother more at risk of risky health behaviour and new pregnancy. While prevention of teenage pregnancy should continue to be a public health priority for governments, measures must be taken at all levels—individuals, communities and governmental—to ensure the continuity of education of adolescent mothers. Several studies have shown that the negative consequences of early childbirth for mothers

and children can be mitigated by further education. Thus, social support programmes targeting adolescent mothers like childcare solutions or alternative education programmes must be developed and implemented by governments or communities, to support teenagers and families struggling to cope with early pregnancy and parenthood.

Abbreviations

ABR: Adolescent birth rate; AOR: Adjusted odd ratio; BMZ: German Federal Ministry of Cooperation; CI: Confidence interval; GIZ: Gesellschaft für Internationale Zusammenarbe; HIV: Human immunodeficiency virus; IQR: Interquartile range; OR: Odd ratio; PGCSS: German-Cameroon Health Programme; RENATA: Reseau National des Tantines; SDO: School dropout; UNFPA: United Nations Population Fund; WHO: World Health Organization.

Acknowledgements

The author would like to thank Drs Andreas Stadler and Gerd Eppel who coordinated the German-Cameroon Health Programme (PGCSS) and gave full support for the implementation of the innovative approach. The authors also thank Georgette Arrey Taku, Executive Secretary of the Network of Aunties as well as all the past executive and through them all Aunties.

Authors' contributions

FN initiated the creation of Renata network. He conceived the study on teenage mothers to describe the social vulnerability of that population. JLST suggested and implemented these analyses on SDO. JLST reviewed the scientific literature and wrote the first draft of the article. LKF contributed to the methods section and reviewed the draft. FN and MTA reviewed the different versions of the draft. All authors read and approved the final manuscript.

Funding

The RENATA network was funded by the German Federal Ministry of Cooperation (BMZ) through the German-Cameroon Health Programme (PGCSS) implemented by Giz. The authors received no funding for the data analysis and paper writing.

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Declarations

Ethics approval and consent to participate

The study protocol was approved by the Ethics Committee of the Catholic University of Central Africa. From 2003 to 2012, verbal consent to participate in the studies were obtained from mothers and written consent was also requested from 2004. Confidentiality was ensured during the data collection process. The dataset used for these analyses was completely anonymous.

Consent for publication

Not applicable.

Competing interests

All named authors declare that they have no competing interest.

Author details

¹RSD Institute, Rue de l'Université, Po Box 7535, Yaoundé, Cameroon. ²Université Catholique d'Afrique Centrale (UCAC), Yaoundé, Cameroon. ³London School of Economics and Political Science, London, UK. ⁴Sorbonne Paris Nord University, Inserm, Inrae, Cnam, Nutritional Epidemiology Research Team (EREN), Epidemiology and Statistics Research Center-University of Paris (CRESS), 93017 Bobigny, France. ⁵Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ) GmbH, Yaoundé, Cameroon.

Received: 4 January 2021 Accepted: 19 December 2021 Published online: 15 January 2022

References

- UNFPA. Girlhood, Not Motherhood Preventing adolescent pregnancy. United Nations Population Fund. 2015. https://www.unfpa.org/sites/default/files/pub-pdf/Girlhood_not_motherhood_final_web.pdf. Accessed 14 Dec 2020.
- ONU. World Population Prospects 2015—Data Booklet (ST/ESA/ SER.A/377). 2015. http://www.un.org/esa/population/publications/ WPP2004/2004Highlights_finalrevised.pdf. Accessed 14 Dec 2020.
- United Nations, Department of Economic and Social Affairs PD. World population prospects 2019: Highlights (ST/ESA/SER.A/423. Vol. 00, Futuribles (Paris, France: 1981). 2019.
- World Health Organization—Department of child and Adolescent health and development. Orientation Programme on Adolescent Health for Health-care Providers. Handout. New modules. Geneva; 2018.
- Alves JGB, Cisneiros RMR, Dutra LPF, Pinto RA. Perinatal characteristics among early (10–14 years old) and late (15–19 years old) pregnant adolescents. BMC Res Notes. 2012;5:531.
- Irvine H, Bradley T, Cupples M, Boohan M. The implications of teenage pregnancy and motherhood for primary health care: unresolved issues. Br J Gen Pract. 1997;47(418):323–6 (PMID: 9219414; PMCID: PMC1313012).
- da Silva AAM, Simões VMF, Barbieri MA, Bettiol H, Lamy-Filho F, Coimbra LC, et al. Young maternal age and preterm birth. Paediatr Perinat Epidemiol. 2003;17(4):332–9.
- 8. Marteleto LJ, Dondero M. Maternal age at first birth and adolescent education in Brazil. Demogr Res. 2013;28:793–820.
- Pogarsky G, Thornberry TP, Lizotte AJ. Developmental outcomes for children of young mothers. J Marriage Fam. 2006;68(2):332–44.
- Turley R. Are children of young mothers disadvantaged because of their mother's age or family background. Child Dev. 2003;74(2):465–74.
- 11. Hofferth S, Reid L. Early childbearing and children's achievement and behavior over time. Perspect Sex Reprod Health. 2002;34:41–9.
- 12. Hoffman SD, Maynard RA. Kids having kids: economic costs and social consequences of teen pregnancy. Urban Institute Press; 2008. p. 460.
- Stevenson W, Maton KI, Teti DM. School importance and dropout among pregnant adolescents. J Adolesc Health. 1998;22:376–82.
- 14. Hofferth S, Reid L, Mott F. The effects of early childbearing on schooling over time. Fam Plann Perspect. 2001;33(6):259–67.
- Lloyd C, Mensch B. Marriage and childbirth as factors in dropping out from school: an analysis of DHS data from sub-Saharan Africa. Popul Stud (Camb). 2008;62(1):1–13.
- Singh S. Adolescent childbearing in developing countries: a global review. Stud Fam Plann. 1998;29(2):117–36.
- Exavery A, Kanté A, Mrema S, Phillips J, Masanja H. Multilevel analysis of childbearing in childhood in Tanzania's Rufiji District. Matern Child Health J. 2016;20(2):447–55.
- Magadi M. Multilevel determinants of teenage childbearing in sub-Saharan Africa in the context of HIV/AIDS. Health Place. 2017;46:37–48.
- Herrera Almanza C, Sahn D. Early childbearing, school attainment, and cognitive skills: evidence from Madagascar. Demography. 2018;55(2):643–68.
- 20. Oronsaye A, Ogbeide O, Unuigbe E. Pregnancy among schoolgirls in Nigeria. Int J Gynaecol Obstet. 1982;20(5):409–12.
- Ferré C. Age at first child: does education delay fertility timing? The case
 of Kenya. World Bank Policy Research Working Paper No. 4833. Policy
 Research Working Papers. 2009; (February): 1–45. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1344718. Accessed 4 Dec 2020.
- Timæus IM, Moultrie TA. Teenage childbearing and dropout from school in South Africa: Evidence from a household panel. https://iussp.org/sites/ default/files/event_call_for_papers/TeenageChildbearing Busan.pdf. Accessed 4 Dec 2020.
- 23. National Institute of Statistics. Second survey of employment and the informal sector in Cameroon. 2011.
- 24. Meadows M, Sadler L, Delvento RG. School-based support for urban adolescent mothers. J Pediatr Health Care. 2000;14:221–7.

- Sullivan K, Clark J, Castrucci B, Samsel R, Fonseca V, Garcia I. Continuing education mitigates the negative consequences of adolescent childbearing. Matern Child Health J. 2011;15(3):360–6.
- Perper K, Peterson K, Manlove J. Fact sheet. Diploma attainment among teen mothers teen birth (age 19 or younger). Washington DC: Child Trends; 2010.
- International Health, GIZ. Réseau national des associations de tantines (RENATA), Cameroun. 2012;2011(sur 187):2–5.
- 28. Institut National de la Statistique. Enquête par grappes à indicateurs multiples (MICS5), Rapport Final. Cameroun. Yaounde; 2015.
- Plan International. Parce que je suis une fille Rapport Afrique 2012
 Progrès et obstacles à l'éducation des filles en Afrique. 2012. https://resourcecentre.savethechildren.net/node/6923/pdf/6921_0.pdf. Accessed 4
 Dec 2020.
- Barnet B, Arroyo C, Devoe M, Duggan AK. Reduced school dropout rates among adolescent mothers receiving school-based prenatal care. Arch Pediatr Adolesc Med. 2004;158(3):262–8.
- Marcotte DE. High school dropout and teen childbearing. Econ Educ Rev. 2011;34:258–68.
- Grant M, Hallman K. Pregnancy-related school dropout and prior school performance in KwaZulu-Natal South Africa. Stud Fam Plann. 2008;39(4):369–82.
- 33. Clark S, Mathur R. Dating, sex, and schooling in urban Kenya. Stud Fam Plann. 2012;43:161–74.
- Marteleto L, Lam D, Ranchhod V. Sexual behavior, pregnancy, and schooling among young people in urban South Africa. Stud Fam Plann. 2008;39:351–68
- 35. Human Rights Watch. Ne marginaliser aucune fille en Afrique: Discrimination dans l'éducation contre les filles enceintes. 2018.
- Chigona A, Chetty R. Teen mothers and schooling: lacunae and challenges. S Afr J Educ. 2008;28(2):261–81.
- 37. Gavin L, Galavotti C, Dube H, McNaghten AD, Murwirwa M, Khan R, St Louis M. Factors associated with HIV infection in adolescent females in Zimbabwe. J Adolesc Health. 2006;39(596):e11-18.
- Stroeken K, Remes P, de Koker P, Michielsen K, van Vossole A, Temmerman M. HIV among out-of-school youth in Eastern and Southern Africa: a review. AIDS Care. 2012;24:186–94.
- 39. Hargreaves J, Morison L, Kim J, Bonell CP, Porter JDH, Watts C, et al. The association between school attendance, HIV infection and sexual behaviour among young people in rural South Africa. J Epidemiol Community Health. 2008;62:113–9.
- Rosenberg M, Pettifor A, Miller WC, Thirumurthy H, Emch M, Afolabi SA, et al. Relationship between school dropout and teen pregnancy among rural South African young women. Int J Epidemiol. 2015;44(3):928–36.
- Hokororo A, Kihunrwa A, Kalluvya S, Changalucha J, Fitzgerald D, Downs JA. Barriers to access reproductive health care for pregnant adolescent girls: a qualitative study in Tanzania. Acta Paediatrica (Oslo, Norway: 1992). 2015;104(12):1291–7.
- 42. Gyan C. The effects of teenage pregnancy on the educational attainment of girls at Chorkor, a suburb of Accra. J Educ Soc Res. 2013;3(3):53.
- Silberschmidt M, Rasch V. Adolescent girls, illegal abortions and "sugardaddies" in Dar es salaam: vulnerable victims and active social agents. Soc Sci Med. 2001;52(12):1815–26.
- Donatus L, Sama DJ, Tsoka-Gwegweni JM, Cumber SN. Factors associated with adolescent school girl's pregnancy in kumbo east health district north west region Cameroon. Pan Afr Med J. 2018;31:1–11.
- Yakubu I, Salisu WJ. Determinants of adolescent pregnancy in sub-Saharan Africa: a systematic review. Reprod Health. 2018;15(1):15. https://doi.org/10.1186/s12978-018-0460-4.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.